\documentclass[11pt]{article}
\usepackage{handout2e}
\begin{document}
\input{macros}
\input{courseMacros}
\thispagestyle{empty}
\setlength{\parindent}{0pt}
\setlength{\parskip}{1.8ex}

\head{CS G714}{Theory of Computation}{Spring 2009}{5 January 2009}{1}

\begin{center}
\Large{\bf Questionnaire}
\end{center}

\medskip

\begin{enumerate}
\item
Have you taken an undergraduate course on the theory of computing?  If
so, what text did you follow?

\vfill

\vfill

\item
Have you taken G713 (Advanced Algorithms) or equivalent?  

\vfill 

\item
Which of the following topics have you covered in detail (say, to the
level covered in Sipser's text) in previous coursework?
\begin{itemize}
\item
Regular languages:

\vfill

\item
Context-free languages:

\vfill

\item
Turing machines:

\vfill

\item
Undecidability

\vfill

\item
NP-completeness

\vfill
\end{itemize}

\item
List any theory topics that are not listed in the schedule that you
would especially like to learn in this course.

\vfill

\vfill

\end{enumerate}
\end{document}

